Author's response to reviews

Title: Over-expressing Akt in T cells to resist tumor immunosuppression and increase anti-tumor activity

Authors:

Yanhong Wu (wuyanhong849@126.com)
Zhenling Deng (dengzhenling1985@126.com)
Yishu Tang (tangyishu111@163.com)
Shuren Zhang (zhangsr@cicams.ac.cn)
Yu-Qian Zhang (yqzhang.cams@gmail.com)

Version: 2

Date: 13 August 2014

Author's response to reviews:

Dear Editor,

We appreciate the prompt comments from Editorial Office regarding to the manuscript “Over-expressing Akt in T cells to resist tumor immunosuppression and increase anti-tumor activity”, and we have made all the changes according to the requests:

1) E-mail addresses of each co-author are listed in the title page
2) Abstract is in a separate page with Methods section added
3) The body of the manuscript is rearranged as instructed
4) Statement regarding to the written informed consent from participants is included (Line 143)
5) Statement detailing compliance with animal care guidelines is included (Line 103)
6) Figures are resized in accordance with BMC’s guidelines

Thank you very much for your attention and consideration.

Sincerely,

Yu-Qian Zhang, Ph.D.
Associate Professor of Department of Immunology
Cancer Institute, Chinese Academy of Medical Sciences
Beijing 100021, China
Email: yqzhang.cams@gmail.com